Author's response to reviews

**Title:** Broad phenotypic spectrum in familial adenomatous polyposis; from early onset and severe phenotypes to late onset of attenuated polyposis with the first manifestation at age 72

**Authors:**

Mef Nilbert (mef.nilbert@med.lu.se)
Ulf Kristoffersson (ulf.kristoffersson@med.lu.se)
Mats Ericsson (mats.ericsson@skane.se)
Oskar Johannsson (oskarjoh@landspitali.is)
Eva Rambech (eva.rambech@med.lu.se)
Peter Mangell (peter.mangell@skane.se)

**Version:** 4  **Date:** 25 November 2008

**Author's response to reviews:**

Minor formatting changes included.